Agata/AMBMin/Sep09/final
Minutes AGATA Management Board 

3rd-4th September 2009, CSNSM Orsay, Start 14:00 on 3rd Finish at 13:00 on 4th.
Present:
D. Bazzacco

A. Gadea

G. Duchêne
E. Farnea
J Nyberg

J. Simpson 


Project Manager and Secretary
P. Reiter
Ch. Theisen
P.J. Nolan


ASC representative

1. Apologies for absence


None
2. Minutes of the last meeting 28th – 29th May 2009 Daresbury Laboratory Agata/AMBMin/May09/02D

The draft minutes Agata/AMBMin/May09/02D were accepted as a correct record of the meeting.

Action JS to distribute as Agata/AMBMin/May09/final.
3. Matters arising from the last meeting and actions outstanding from last phone conference

Travel funds in Orsay. Still an issue since overspends from previous years are carried forward and the debt has not been cleared.

4. ASC Report 
P.J. Nolan reported on the ASC meeting of 16-17th June 2009 in Stockholm 

Items discussed:
MoU signed by all except France, but signature is now expected soon.
Organisation of AGATA and the AMB, project manager, see later.
Funding in each country and status.
Running costs, lack of, see later.
Location of AGATA in the future.
W. Korten appointed as AGATA spokesperson for GSI campaign. To work with M. Bentley who is the currently the PRESPEC fast beam spokesperson.
A workshop to be organised in GSI early next year in time for GSI PAC.
Publication policy. 
What is the AGATA collaboration. Lists are being prepared.
Launch of AGATA. D. Napoli and F Azaiez to organise an inauguration at LNL. Formal inauguration in afternoon, party in evening, collaboration council in following morning.

AGATA weeks.
Turkey physics meeting in April/May 2010.

AGATA input to the new NuPECC Long Range Plan.

A meeting at GSI on 1st July took place between all laboratory directories, PAC chairs and AGATA representatives. It was organised by S. Lenzi.

Agreed overall plan for AGATA. LNL-GSI-Ganil.


Stay in LNL until summer 2011.


Earliest start GSI October 2011.


Earliest start GANIL March 2013.


The meeting was very positive.

Organised campaigns encouraged.

5. Reports from Working Groups

a. Detector module and characterisation

See full report of Peter Reiter, will be distributed with the minutes.
Status of AGATA Detectors, Deliveries and available Detectors 

Detector customer acceptance tests


· Detector A007 : Delivery 28/07/09 – tested at IKP Cologne, not accepted due to core energy resolution, mechanical problem , see detailed report 

· Detector B001 : Delivery 22/01/09 – Rejected 06/02/09 due to leakage current, done at Cologne                                                                                                           

Delivery 18/06/09 – mounted in ATC3, see results below, done at Cologne

· Detector B005 : Delivery 05/03/09  – mounted in ATC3 first for CAT, leakage current in ATC3 , -  2nd CAT test in Daresbury test cryostat, detector is within specification no leakage current, - detector is mounted back in ATC3,  -see results in report, no report sent yet, done at Cologne

· Detector B006 : Delivery 16/07/08 – accepted 17/10/08 – shows leakage current after mounting in ATC3, another test in single test cryostat confirmed result, detector returned to Canberra for repair due to leakage current, done at Cologne  

· Detector  C006 : Delivery 30/04/09 – accepted, mounted in ATC3. Done at Cologne

Future deliveries Information provided by Canberra September 3rd, 2009

Delivery schedule for the AGATA detectors in the coming months (current orders and reparations):

- C003 (in reparation): end of September 2009

- B006 (in reparation): end of October 2009

- B007 (Germany): late October 2009 / early November 2009

- C007 (Germany): early December 2009 

- A008 (Germany): end of January 2010

Please keep in mind that this remains a delivery target and that this can be affected by the quality of the HPGe crystals. 

Concerning the sequence of geometries (A,B or C) to be delivered, we are simply following the alphabetic order for a given series. If you know 2-3 months in advance that you will need a particular geometry to complete one of your triplets, we could provide it in priority.

Status of AGATA triple detectors

ATC1 @ Legnaro 

March: modification of warm preamp grounding removed oscillation at 40 MHz 

April: modification of cold grounding removed oscillation between 1 and 20 MHz

After modification in cold part:

- swapped segments green B4 &B5

- return current wires were shielded 

- Shield of pulser cable grounded

- ground layers of cold PCB connected

- PT100 disconnected from cold finger

June: Repair of 2 broken segments – Blue D2 and E6

Two segments got lost after cooling the detector – one after mounting the detector in frame (both on blue detector) F4 and C2

Detector in demonstrator frame – operational besides the 2 missing segments

ATC2: 

HF-oscillations removed since modification of grounding of micro match connectors and cold grounding in Cologne in March 09.  Good performance and energy resolution as seen in figures below:

Crosstalk shows expected contribution at 10-3 level.

ATC2 delivered to Legnaro in July 2009, tested after delivery - similar performance but one segment is missing (Green A5 – broken FET or signal wire)

Detector mounted in demonstrator frame – operational besides the missing segment

June: micro match task force with Alberto & George -> new connectors found – new warm PCB to be designed and printed – still in progress

ATC3: 

A002, B001 & C006 mounted 

– B001 is mounted for acceptance test in ATC

Core energy resolution values at 1.3 MeV are typically 200-300 eV above test cryostat results:

Red:  2,39 keV

Green: 2,52 keV

Blue:   2,50 keV

Status of segments of B001 

Segment B6: 1,11 keV at 3 µs shaping time and 1,39 keV at 6 µs – small indication of leakage current

AXIS test on Core signal: 
	AXIS without filter:

FWHM @ 1.3MeV

Red: 3.11 / 3.15keV

Green: 3.14 / 3.23keV

Blue: 3.13 / 3.38keV


	AXIS with filter:

FWHM @ 1.3MeV

Red: 2.27 / 2.4keV

Green: 2.36 / 2.43keV

Blue: 2.3 / 2.37keV




AXIS with filter good performance – FWHM and FFT spectra OK

New set of HV boxes test:

1st problem: spikes (mainly on third box - 127)

2nd problem: increasing current while increasing the voltage (seen on boxes 1 & 3 - 147 & 127) and constant current above a certain voltage

   box 1 (147) U=4000 V  I=500 nA  - too high

   box 2 (126) U=4000 V  I=-135 nA - OK

   box 2 (127) U=4000 V  I=1133 nA - too high baselines are floating with box 1 and 2 – these boxes are not operational

3rd problem: core signal after the main amplifier shows oscillation with 50 kHz and  25 mV amplitude.

ATC3 was used for additional R&D related to (i)  new cables in cold part and (ii) new capacitive LN2 readout , see details below

Scanning

Liverpool AGATA detector characterisation status report (Sept 09)

The repair to AGATA C001 was completed and the detector operational from 6th June 09. A number of troubleshooting steps were followed with advice from the Cologne group and a large amount of experience has been built up in Liverpool. The performance of the detector was verified prior to recommencing the scan.

A brief summary of the status: 

Singles and singles vs. HV scans (@2000, 3000, 4000 and 4500V) recorded. [COMPLETE] 

Coincidence scan completed with Caen HV power supply (finished 1 September). [COMPLETE] 

Further singles HV scans now in progress (est 7 days) 

Test new scanning method (est 14 days) 

Singles side scan with 137Cs and 241Am (est 14 days)

 Completion by mid October with contingency

We will continue to collect more coincidence data if a new detector is not available on this timescale. A conversion code has been written to convert the Liverpool data format to root. Data will shortly be made available to the collaboration.

New Publications: 

Dimmock et al., IEEE TRANSACTIONS ON NUCLEAR SCIENCE, VOL. 56, NO. 3, JUNE 2009 

Dimmock et al., IEEE TRANSACTIONS ON NUCLEAR SCIENCE, VOL. 56, NO. 4, AUGUST 2009

See detailed report from A. Boston with experimental results, which will be distributed with the minutes. 
Activities at Orsay:

· A005 or A006 are tested and ready to be mounted into test cryostat for scanning at Orsay. Mounting of A-type detector into test cryostat postponed due to temporarily missing expertise and manpower at Saclay. 

· Plan: Detectors and cryostat will be shipped from Saclay to Cologne at September 15 for mounting A detector. 

· To be discussed within AMB, priority between (i) scanning at Liverpool, (ii) scanning at  Orsay, (iii) assembly of ATC4.
Priorities of detector and characterization group

Available detectors for scanning and fourth triple cluster detector ATC4

A005, A006 (at Saclay, available)

B005 (at Cologne, available)

C001 (at Liverpool, available October 09)

Option A
· Scanning of A005 at Orsay, implies refurbishing of scanning cryostat, new cryostat hardware would allow standard AGATA configuration

· Scanning of B005 at Liverpool, after mounting in Liverpool 

· A006 and C001 available for ATC4, missing B detector, repaired B006, new B007

  Option B
· Scanning of A005 at Liverpool, after mounting in Liverpool 

· refurbishing of  Orsay scanning cryostat for next B detector, new cryostat hardware would allow standard AGATA configuration, scanning of B detector later, either repaired B006 or new B007

· assembly of ATC4 with A006, B005, C001 after end of scanning at Liverpool September 2009 

The AMB agreed to proceed with Option B.
Repair of broken segments of ATC1 and ATC2 at Legnaro

Proposal: results of latest developments related to (i) new cabling and (ii) modified pre amp connectors will be included in next repair at Legnaro. Repair should include exchange of cold cabling of all detectors, replacement of pre amp micro match connectors  and new warm PCBs in warm part of triple cryostat. 

Other point raised in discussion.

Turkish detectors and digitizers have been shipped from Ankara now on way to GSI.
Potentially delivery another ATC by end of December, 4th in January. 

Action PR to urge CTT to work in parallel to mount the 4th.

Planning to have 4 ATC’s operating in January in LNL. 

Offline discussion regarding allocation of detector tasks to Liverpool and Saclay to help the pressures at Cologne.

Action PR, CT and PJN to discuss.

All pre-amplifiers needed for 4 ATC’s are ordered.

Try to cover cryostat costs using the running costs. Funds need to be transferred from Cologne. Order to be placed by EF to CTT.

Action EF and PR.

b. Front end processing
Presentation by Dino will be distributed with the minutes.

Summary of status:

	Digitizers

6 in use

3 sent to LNL (yesterday)

ATCA Carriers

12 in use (6 Padova, 6 IPNO) 2 per crystal required

6 (IPNO) arrived qt LNL on 1/9

7 under test at IPNO (delivered last week)

10 to be repaired at manufacturer

2 not yet delivered

TCLK cards

6 in use + 3 delivered

7 ready in IPNO

Processing mezzanines

42 in use in LNL

8 available at LNL

49 available at CSNSM

15 delivered this week

2 in repair


	GTS Mezzanine


5 prototypes in use


4 series production in use


16 series production tested OK


10 more to be produced

Trigger processor


In use (based on GTS mini-tree + evaluation board)

PCIExpress


12 LINCO-1 (2/ detector) prototype OK


2 more available


2 LINCO-2 (det/card) proto OK


License issue

ATCA crates


2 in use at LNL


2 available (LNL + Orsay)

Optical Fibres


In place for 6 Clusters (digi-prepr.)

In repair (PCIe + ancillaries)



A plan for the PCIExpress licence and ordering digitizers was agreed.

Action DB and PR.
Joel Chavas will leave by mid-November. This will affect GTS development.
Status of firmware and software:

· Digitizers

· Mainly done

· Implementation of ToT processing to be scheduled 

· ATCA carrier board

· Well advanced, including PCIe readout 

· Processing mezzanines

· Basic implementation done

· Various developments needed to comply with specs

· GTS

· First mini-tree in use; to be debugged

· Global trigger 

· Tests of timestamp-based implementation started last week
ToDo’s firmware and software:
· Remove know bugs

· Reduction of event dead time from 25 ms to ~1 ms

· Implementation of per-channel setup parameters

· Testing the exponential-average self-triggered blr

· Reduce number of readout samples from 160 to 80

· Verify if possible to use CFD keeping low thresholds 

· Improve diagnostics

· Counters of local triggers, validations, rejections, …
· Readout of long traces

· 100 ks if one channel

· 10-20 ks if 6 channels in parallel
Status of Orsay (CSNSM) with a full system as a test bench

· 1 detector 

· 1 digitiser + PS + cooling + fibres

· 1 pair of ATCA carriers

· 7 processing mezzanines

· 1 GTS mezzanine
· PCIe readout (1 LINKO)
· DAQ
System basically ready. To be implemented in scanning table area.


Local data processing

· Pre-processing

· Calibrations and cross-talk correction for energy OK

· Cross-talk correction for traces still to be done

· Timing ( to be improved

· PSA

· Grid search

· Full search ~300 evts/s/core

· Adaptive search ~5 times faster with the same performance (from week27 experiment)

· Other algorithms?

· Improvement of signal bases?

· When to stop writing raw data

· Group needs to be more proactive. Need to consider how this work is managed in the new structure.
c. Data Acquisition
Data Flow
· Data Flow meeting week 23 (Mainly ADF)

· ADF

· Bugs fixed, new features (vertex frame), … 

· Need optimization

· July beam test 

· ADF inconsistencies

· EB issues

· Watchers (spy in progress, very root orientated)

· Topology generator (Xavier). 

· AOB like narval nodes reboot, … 

· Need a expert meeting at Legnaro before next beam time (and still need more coordination).
Administration
· Ticket system

· Visit of Yann + Eric at Legnaro on week 26. 

· Faster link between box and T2 room

· Very long todo list. In progress but.

· KVM

· 3rd Visu

· Faster I/O (GPFS ? CNFS, TCP Sockets ?).

· Disk privileges : urgent.

· Backup system

· User accounts, LDAP server

· 3rd DS, SRM (Grid)

· 2nd DNS zone

· …

· Need already some hardware maintenance?

· Narvalds2 DS

· Kernel panic on some nodes

Services
· Long todo list

· Run Control

· Need coherent implementation (eg process_stop) on actor side

· “Clean” destroy and “Ultimate reset” (not only RC stuff)

· Use run number

· Log implementation in actors

· Save used configuration (topology, config, algos, …).

· GUI : lot needed, lot of requests (Jurek ????)

· … 

· “Friendly” log display. To be tested.
Hardware and funds
· Present hardware enough for 3 TCs

· Still fund issues for travel? Already discussed.
· Reminder for the full (6 TC) demonstrator

· 9 (+1) pizza : ~ 18 (+2) kEuros ( Italy if no PCI

(Need now one more for new SRM)
Action DB and CT
· Need 4 FC interfaces (2-4 kEuros). ( Italy if no PCI
Action DB and CT
· 60 TB disk (Sun Storage) : 40-60 kEuros ( Germany OK if no PCI licence
Action PR and CT
· Adacore license ? (IPNO ?)

· DA workstations ? ( visu 4-5. Administration like visu 1-3.

Slow Control
· SC meeting week 25

· GSC Draft specifications distributed (Luz Guevara et al.)

· Need feedback

· Need face to face meeting ~ October

Documentation

· Wiki filled slowly. 

Remarks
· Very long todo list, action list growing.

· Not that easy to push teams although a lot of meetings (17 this year).

· Feedback needed, but only from very few peoples (DB, DM, …).
d. Ancillary detectors and ancillary detector integration
Already commented in the last Telephone Conference

6 new AGAVA modules with modification on the Ethernet port and NIM input have been received for testing. With these, a total of 7 modules have been delivered.

Joel Chavas has upgraded the firmware (GTS-2) of the 6 cards and tested the Ethernet interface, 5 modules have successfully past the Ethernet connection check. The second part of the test was the functional test with the Milano DAQ by Sergio Brambilla, During this test it was checked that the "Trigger Request" NIM input problem has been repaired and now it can be piloted by standard NIM signals. Nevertheless, checking the output Local Trigger and Validation NIM signal from the front panel, it was observed a double signal not present in the first delivered module. The modifications on the NIM buffer impedances are being checked. 
The second action was a test of the first part of the readout for PRISMA/DANTE and for HELENA complementary detectors. Both front end where performing the readout of VME ADCs, TDCs and Scalers.  Both front-end use a TCP client to deliver the buffers to the NARVAL PRODUCER. The NARVAL producer sent the buffers to the NARVAL ACTOR called DATA_RATE. The test was performed by Enrico Calore and Daniele Mengoni taking care of the NARVAL part and by Nicola Toniolo and Sergio Brambilla for the front end of PRISMA/DANTE and HELENA respectively.
The transfer rates obtained for the PRISMA/DANTE readout had been of the order of 5MB/s corresponding to about 34 kHz event rate.
The HELENA readout system is programmed in JAVA and is suffering from problems to keep the client opened, nevertheless the transfer rates are of the order of 2.5 MB/s corresponding to about 10 kHz event rate.
The next step in September will be to build a NARVAL ACTOR that prepares the events to be sent to the NARVAL MERGER, adding the Keys to the event to build the complementary detector data frame.

New issues:

Roberto Isocrate (PD) has identified the problem with the AGAVA NIM outputs, during the mounting there was a change in one of the integrated components (inverter) between the first production module delivered and the rest. The integrated components will be changed as soon as we purchase the new ones.
Next DAQ test should include the delivery of the data buffers to the Ancillary Merging NARVAL actor. To be performed before the next in-beam test.

The engineers at GANIL and GSI have required a minimum GTS system to check the clock readout and merging. The system is composed of:


-Two AGAVA cards complete with GTS mezzanine on-board.

-One VME carrier with a GTS mezzanine acting as clock generator 

The amount of GTS mezzanines produced is not sufficient to have the AGATA Demonstrator and a number of GTS mezzanines for testing purposes at the host labs. Funds foreseen for the production of GTS mezzanines for AGAVA in the project capital plan have been invested in the design and production of the carrier board by INFN.
 Since the design of the GTS mezzanine is frozen, possible the production of extra mezzanines that will be used latter in the next phases of AGATA. The cost of the production of 10 mezzanines is about 20k€. 

Possibility to anticipate funds from phase 1? Request from GSI. 
Action AG
Possibility to produce and test the cards in reasonable time?

Action AG to plan and find the effort.
Discussions on how to couple the GSI/PRESPEC detectors to AGATA ongoing, documents with proposal will be distributed for discussion very soon.

e. Design and Infrastructure

Last VC took place on Sept 02, 2009

ATC#2: 
The TC is on the array in the full configuration:

· powered by the AXIS LV PS 

· preamps LVs continuously adjusted via 15m sense-wire cables; external LV filter used in addition with AXIS (later to be incorporated in patch box)

· HV boxes operational

· digitisers powered by the AXIS module

· a standalone PLC

Preliminary resolutions are satisfactory. Detailed measurements are needed.

ATC patch boxes: Work and discussions between Saclay, CTT and Koeln on-going on the patch boxes. EMC to be carefully studied.

LV modules (AXIS): Two AXIS modules upgraded with switches to be able to operate in standalone mode have been delivered in Saclay. They are tested and validated.

To be installed in LNL in week 40 (Sept 28-Oct 2). Sets of cables for 5 TCs, Profibus network and last PLC process version will also be implemented.

Cabling: Installation will include extra cabling which should be mounted efficiently onto array, possible issues with cable management.
To be chased up.
HV module: Problems in Koeln encountered with the 3 HV boxes for ATC#3. Two cannot rise the voltage at nominal value and the third unit generates 50 kHz oscillations (FWHM degraded by about 400 eV). GSI and Saclay to move to Koeln for inspection and tests, likely on week 37 (Sept 7-11). To be confirmed.

On-going profibus tests at Saclay.

Second revision HV-box drawings done at GSI, agreed by Daresbury. 9 boxes to be produced for  end October.

Autofill: A full autofill system mounted at GSI for VCC tests. Inconsistencies already found. VCC possibly ready late next week for installation at LNL on week 38 or 41, to be confirmed.

GUI: OPC version almost finished. Ready for installation by week 40 (see LV module paragraph).

Grounding: Grounding expertise held on July 8 at LNL with Messrs Charoy, Richard, Karkour and Tripon (in view of the AGATA installation at GANIL). Preamps, MDR cable, electronics racks and mechanical structure groundings have been tested. Preliminary report distributed. EMC can be greatly improved. 

Presentation by N. Karkour on the draft EMC and Grounding report, distributed to the AMB prior to the meeting. Final paper will be distributed next week. The paper has several recommendations that will improve the system.

The EMC issues were discussed after Nabil's presentation. 
AMB recommends the following.

For the AGATA Demonstrator at LNL we should make use of an EMC expert (Nabil) willing to help and give advice concerning grounding problems. The AMB proposes that Nabil gets involved in the different teams (preamp, infrastructure, etc) where grounding is important. Grounding problems at high frequency should be solved as much as possible, following the advice of Nabil, without delaying the AD project at LNL.

For the next phases of AGATA (GSI, GANIL) it makes sense to have a working group for the general performance of the array and this should include the EMC experts. To create such a group is a task for the new AMB structure.

f. Gamma-ray tracking, simulations and data analysis

Simulations: EF says code is up to date. Some work on GSI set-up by Cesar Pardo.

Tracking:
- No updates. Waiting for a source run with the demonstrator to validate
  "real" tracking parameters. 

Action DB will inform Waely where the data is.

Data bases:
Work in Lyon on hold.

Data analysis:
- Two workstations delivered to LNL end July. Discussion with Olivier and DAQ group of OS installation and system administration of these machines
- Status of the analyses of March and June/July in-beam commissioning runs, and source runs. 
EF reported, presentation will be circulated.
Summary of “second” in-beam test:
· Parasitic beam time (end of May) suffered several problems, repeated end of June

· 56Fe at 220 MeV ( 197Au

· (Similar to the 56Fe on 208Pb performed in 2001 with MARS)

· Scattered ions detected in the MCP DANTE, taken with the AGATA electronics (AGAVA not yet ready)

· Ge-Dante coincidence selected by the AGATA real-time digital trigger

· Analysis of ATC1 done in real time in the Narval farm
· Analysis of DANTE done off-line after the test as the online processing programs were not ready

· Preliminary results (obtained in just one week) look very promising and there is enough statistics for detailed analyses

Week 12 analysis close to end.

Action JN to call a meeting of data flow and analysis.
6. Installation of the demonstrator at LNL and project planning
PAC Results
· 13 days of in-beam commissioning awarded (to be split into 3runs as discussed in previous VC)

· 3 experiments approved (17 days in total)

· Should produce a document detailing our plans for the commissioning; should then present our results at the next PAC. February 2010. Expect lots of proposals then.
· Tandem beams from 12 Oct to 8 Dec 8 ALPI beams from 1 Feb to 8 March (rough schedule)
· Problem: most experiments of next semester will be in AGATA experimental hall, almost Impossible to work in the area.

ATC@LNL

· ATC2 was delivered to LNL second half of July

· Temporary solution for DSS implemented

· ATC1: LV, HV digitizers powered with standard solutions

· ATC2 LV, HV digitizers powered with AXIS module, second PLC to drive the HV

· Both ATCs kept cold with autofill without problems (7 hour cycle)

· Mechanics, local engineers trained by John Strachan. Still some alignment and mounting issues to be solved.
· Electronics to acquire both ATCs set-up in early August
Action: EF to contact John Strachan regarding detector transportation with the laboratory.

Discussion of an offline run with Cf source.

Need to find a sensible slot.

Action EF and JN

7. Funding status, capital allocation and running costs

Agreed spending plan again. See items listed in last minutes.
Action PR to spend up (see items above and action in 5c)

Germany to purchase digitizer, mechanical item, remaining funds on disks.

Agreed plan for pre-processing for Turkey. Turkey order pre-processing items from GSI (letter of credit could not be organised in France). France will sell items to GSI.

Action GD

Italy to buy pizza + FC interfaces if have funds released from PCIExpress licence.

Action DB (and action in 5c)

Outstanding items not allocated
20 k € GRID Bologna

20 k € GRID Lyon

21 k € Adacore licence (IPNO?)
20k euro GTS for AGAVA

8. Next AGATA week and physics meetings
AGATA @ LNL Physics Workshop issues. To be organized by A Gadea and S Freeman. To discuss LoIs and submissions for following LNL PACs (June 2010, January 2011) 11-14th May 2010.
Two possible locations are considered:

-Istanbul

-Conference location which belongs to Ankara Univ. located at the historic town of Side by Antalya. 
The town is far from main flight destinations in Turkey (Istanbul and Ankara)

Ankara University has agreed to organize the meeting, 17k€ available for the organization

Local organizing committee:

Ayse Atac, Ayse Kaskas from AU
Nizamettin Erduran from IU
Sefa Erturk from Nigde Univ
+Secretary and young collaborators to be defined
AMB discussed wider physics programme that could be included.

Physics at GSI meeting to be organized by W Korten and M Bentley, early 2010.

Action JS to find out what the plan is.


AGATA Week.



LNL 20-22nd January 2010. EF to check not a show stopper.


Sessions on GSI to be included.

Action: JS to find out what the status is at GSI, meetings, tests etc.
9. Funding, capital allocation and running costs.


Running cost spends at LNL to date 2009, grand total circa 27.4 k €
See lists in document distributed by email.
Need running costs actions for ASC to be completed

Action JS and EF to discuss. 
(Note from discussions in the detector section need 3.5k € mechanics, 7k € for preamplifiers is required.)
All costs to include tax.

10. AGATA Organisation

PJN circulated a paper for comments on the role of the AGATA project manager.

Comments to PJN by 28th September.
11. AOB

Further  grounding discussion with N Karkour.

For discussion NK to use ANB as primary contacts.
NK will take MDR cable from Strasbourg and make modifications.

NK will make a modification on one pre-amplifier in consultation with PR et al.

Thanks to N. Karkour for hosting the meeting.
13. Date and location of next meeting and dates of phone conferences.



12th to 13th November 2009 LNL, lunchtime to lunchtime meeting.
Phone conferences / Lyon system (GD to organise)

16th September, 10.15 CET

2nd October, 10.15 CET

16th October, 10.15 CET

28th October, 10.15 CET










